Author's response to reviews

Title: Tumour-associated endothelial-FAK correlated with molecular sub-type and prognostic factors in invasive breast cancer

Authors:

Annika Alexopoulou (a.alexopoulou@qmul.ac.uk)
Colan M Ho-Yen (c.ho-yen@qmul.ac.uk)
Vassilis Papalazarou (vasilispapalazarou@gmail.com)
George Elia (g.elia@qmul.ac.uk)
Louise J Jones (l.j.jones@qmul.ac.uk)
Kairbaan Hodivala-Dilke (k.hodivala-dilke@qmul.ac.uk)

Version: 3
Date: 20 March 2014

Author's response to reviews:

Dear Sir/Madam,

Thank you for pointing out the details that were lacking in our revised manuscript. We have addressed these points by:

1) Adding in an acknowledgements section

2) Adding in the statement "This study followed REMARK guidelines for tumour marker prognostic studies" to the methods section, along with the appropriate reference.

We hope this meets the journal requirements. Please let me know if there are any other queries.

Yours sincerely,

Colan Ho-Yen